Author's response to reviews

Title: Rate of Intraoperative Problems during Sacroiliac Screw Removal: Expect the Unexpected

Authors:

Georg Osterhoff (georg.osterhoff@medizin.uni-leipzig.de)
Jonas Noser (jonas.noser@uzh.ch)
Kai Sprengel (kai.sprengel@usz.ch)
Hans-Peter Simmen (hans-peter.simmen@usz.ch)
Clément Werner (clement.werner@hirslanden.ch)

Version: 2 Date: 28 Mar 2019

Author’s response to reviews:

Dear Dr Barling

Please find enclosed a revised version of our manuscript entitled “Rate of Intraoperative Problems during Sacroiliac Screw Removal: Expect the Unexpected”, which we had submitted as an original paper for publication in BMC Surgery.

We made the changes suggested and hope that the manuscript now meets the standards for publication in BMC Surgery.

All co-authors have seen and agreed with the revised version of the manuscript.

Yours sincerely,

Georg Osterhoff